Author's response to reviews

Title: Physical activity patterns in older men and women in Germany: a cross-sectional study

Authors:

Anna Moschny (anna.moschny@rub.de)
Petra Platen (petra.platen@rub.de)
Renate Klaassen-Mielke (renate.klaassen-mielke@rub.de)
Ulrike Trampisch (ulrike.trampisch@rub.de)
Timo Hinrichs (timo.hinrichs@rub.de)

Version: 5 Date: 12 July 2011

Author's response to reviews:

Dear BMC Series Editorial Production Team,

there were some last minor errors in the manuscript:

1) The manuscript is written in American English. In some cases I had to adapt the words "analyse" and "behaviour" to the American English writing.

2) Table 1 contained a comma instead of a spot in the column of p-values.

3) Table 2: the term "interview date" was changed to season as it is used in table 1.

Kind regards,

Anna Moschny

in behalf of all authors